Author’s response to reviews

Title: The health conditions of inmates in Italy

Authors:

Fabio Voller (fabio.voller@ars.toscana.it)
Caterina Silvestri (caterina.silvestri@ars.toscana.it)
Gianrocco Martino (gianrocco.martino@gmail.com)
Eleonora Fanti (eleonora.fanti@ars.toscana.it)
Giorgio Bazzerla (gbazzerla@ulss.tv.it)
Fabio Ferrari (f.ferrari@asl5.liguria.it)
Marco Grignani (marco.grignani@uslumbria1.it)
Sandro Libianchi (sandrolibianchi@hotmail.com)
Antonio Maria Pagano (a.pagano@aslsalerno.it)
Franco Scarpa (f.scarpa@usl11.toscana.it)
Cristina Stasi (cristina.stasi@gmail.com)
Teresa Di Fiandra (t.difiandra@sanita.it)

Version: 2 Date: 19 Sep 2016

Author’s response to reviews:

Point-by-point response to the Editor’s and reviewers’ comments

We thank the Editor and the reviewers for giving us an opportunity to revise our manuscript.

Please find enclosed the revised manuscript.

The manuscript has been improved according to the suggestions of reviewers.

All changes made in the revised version are in bold text.

Editor:
Declarations

We have added the declarations, line 3, pg. 13

- Ethics approval and consent to participate

We have added this section, line 5, pg.13. We have also have specified the reasons of the assertion of this section in the Methods section lines 5-12, pg. 6

- Consent to publish

We have added this section, line 7, pg.13

- Availability of data and materials

We have added this section, line 9, pg. 13

- Competing interests

We have added this section, line 11, pg. 13

- Funding

We have added this section, line 13, pg.13

- Authors' Contributions

We have added this section, line 16, pg.13

- Acknowledgements

We have added this section, line 24, pg.13

Authors' Information

We have added this section, line 5, pg.14